Author’s response to reviews

Title: Clinical outcomes of children with acute asthma and pneumonia in Mulago hospital, Uganda: a prospective study

Authors:

Rebecca Nantanda (rnantanda@gmail.com)
James K Tumwine (kabaleimc@gmail.com)
Marianne S Ostergaard (moster@sund.ku.dk)
Grace Ndeezi (gndeezi@gmail.com)

Version: 4 Date: 17 September 2014

Author’s response to reviews: see over
Cover letter to Editor

Research article MS: 1351790684125643

Title: Clinical outcomes of children with acute asthma and pneumonia in Mulago hospital, Uganda: a prospective study

Dear Editor,

We extend our appreciation to you and your team for the comments on this manuscript and accepting it for publication in this prestigious journal.

The manuscript has been revised as follows;

1. The information on funding has been included in the acknowledgement section.

2. This study conforms to the STROBE guidelines and a statement to this effect has been included in the methods section.

3. The manuscript has also been formatted to adhere to the journal guidelines.

The revised manuscript with track changes in MS Word programme is provided.

Thank you so much

Yours Sincerely,

Rebecca Nantanda
Corresponding author